Author's response to reviews

Title: Ankle Blood Pressure as a Predictor of Total and Cardiovascular Mortality

Authors:

Heikki Hietanen (heikki.hietanen@fimnet.fi)
Rauni Paakkonen (rauni.paakkonen@ktl.fi)
Veikko Salomaa (veikko.salomaa@ktl.fi)

Version: 3 Date: 8 February 2008

Author's response to reviews:

Dear BioMed Central Editorial production Team,

Thank you for the list of formatting changes requested in our manuscript (your e-mail of Feb 06, 2006). Our responses are as follows:

Title - "Please use sentence case in the title". Corrected as requested.

Competing interests: The expression is corrected as requested.

Authors contributions: corrected as requested.

Table title and legend. "The table title should appear above the table. The legend (or key) should appear below the table". We believe that this was already the case. We did not find anything to correct with regard this point.

References: The numbering of references 5, 6, and 7 is corrected and the reference 31 is now cited in the text.

We hope that everything is now clear. However, if you have any other questions please do not hesitate to contact me again.

Yours sincerely,

Veikko Salomaa, MD, PhD
Research Professor
KTL-National Public Health Institute
Mannerheimintie 166
FI-00300 Helsinki
Finland
E-mail: veikko.salomaa@ktl.fi
Phone: +358-9-4744 8620